Author's response to reviews

Title: A comparison of low-dose risperidone to paroxetine in the treatment of panic attacks: a randomized single-blind study.

Authors:

James M Prosser (jprosser@chpnet.org)
Samantha Yard (samyard@u.washington.edu)
Lisa J Cohen (lcohen@chpnet.org)
Igor I Galynker (igalynker@chpnet.org)

Version: 2 Date: 20 August 2008

Author's response to reviews:

James Prosser, MD
Dept. of Psychiatry
Beth Israel Medical Center
First Ave. at 16th St.
New York, NY. 10003
212-844-1422
jprosser@chpnet.org


Dear Sirs:

I am submitting for consideration of publication in your journal BMC Psychiatry our original research article entitled “A comparison of low-dose risperidone to paroxetine in the treatment of panic attacks: a randomized single-blind study”. We feel this is an important body of work that will be of interest and appreciated by your readership.

I am the contact author for this work and I can be contacted at the address, e-mail, and telephone listed above. Please do not hesitate to contact me if you have difficulty with the electronic files and their upload. I am glad to provide whatever help I can. I am happy to address any other concerns or questions you might have.

Thank you for considering our article, and thank you for your time on my behalf

James Prosser, MD